Dear editor,

We are resubmitting the Manuscript ID 69708 entitled "Flip-over of blood vessel intima caused by vascular closure device: a case report" to "World Journal of Clinical Cases". Our responses to the comments by the reviewers are outlined below. Please also see the revised manuscript for details. For easily reading, we use the TRACK function of MICROSOFT WORD. If you do not like the tracks, please simply select “Accept changes” to get rid of the TRACK markers.

Reviewer comments:
Reviewer #1:
Scientific Quality: Grade C (Good)
Language Quality: Grade A (Priority publishing)
Conclusion: Minor revision
Specific Comments to Authors: I appreciate the author's efforts in writing the exciting case report. Figures were of good quality, and legends were well explained. Consider including other articles that talked about other complications secondary to starClose and VCD. Another case series mentioned the thrombus secondary to the starClose device. DOI: https://doi.org/10.3174/ajnr.A2001 It is prudent to include other complications secondary to VCDs https://doi.org/10.1053/j.tvir.2015.04.008 this article talked about other complications secondary to VCDs. Rodriguez et al also reported pseudoaneurysm as a complication. Authors talked about more and less common complications secondary to the starClose device, and readers would also get benefit if authors include other complications secondary to the starClose device. If authors include a table containing complications secondary to the starClose device would increase the quality of the manuscript. Thank you for the opportunity.
Response: Thanks for the comment. As suggested, we have included articles that talked about complications secondary to starClose. The above mentioned references have been cited and discussed. Accordingly, we have added a table (Table 1) containing complications secondary to the starClose device. Please check Table 1 and the revised manuscript for details.

EDITORIAL OFFICE’S COMMENTS

Authors must revise the manuscript according to the Editorial Office’s comments and suggestions, which are listed below:

(1) Science editor:
The manuscript reported a case of flip-over of blood vessel intima caused by vascular closure device, the author's work is interesting, I think it is a good idea to illustrate other complications of starClose. The author should apply the updated literature in nearly three years.
Response: Thanks for the comment. As suggested, we have added a table (Table 1) containing complications secondary to the starClose device. Additionally, we have updated the references. Please check Table 1 and the revised manuscript for details.

(2) Company editor-in-chief:
I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. Please upload the approved grant application form(s) or funding agency copy of any approval document(s).
Response: Thanks for the comment. As suggested, we have provided the original figure documents in PowerPoint. Additionally, we have uploaded the approved grant application forms.